Author's response to reviews

Title: The in vitro antisickling and antioxidant effects of aqueous extracts Zanthoxyllum heitzii on sickle cell disorder

Authors:

Nanfack P Nanfack NP (pauline600@yahoo.fr)
Prosper N Biapa BNPC (brbiapa@yahoo.fr)
Constant A Pieme PCA (apieme@yahoo.fr)
Jocelyne V Ama AMVJ (movicky@yahoo.fr)
Bruno M Moukette MMB (mouket2006@yahoo.fr)
Jeanne Y Ngogang NYJ (jngogang@yahoo.fr)

Version: 3 Date: 21 June 2013

Author's response to reviews: see over
Cover letter with reviewers and editor`s comments

By PIEME constant Anatole

21th June 2013

1- Editors comments

Question: We notice that that you refer to sickle cell disorder as primarily afflicting 'the Black race'. Please note that we do not consider this term to be appropriate.

…….. characterized by red blood cells that assume an abnormal, rigid, sickle shape and known to be one of the diseases afflicting primarily the population living mostly in Africa, South America and Asia.

Answer:

…….. characterized by red blood cells that assume an abnormal, rigid, sickle shape and known to be one of the diseases afflicting the population living mostly in Africa, South America and Asia.

Thanks